Title: Effects of a community-based healthy heart program on increasing healthy women's physical activity: a randomized controlled trial guided by Community-based Participatory Research (CBPR)

Authors:

Raha Pazoki (pazkipg@bpums.ac.ir)
Iraj Nabipour (nabipourpg@bpums.ac.ir)
Nasrin Seyednezami (Nasrin_naz2001@yahoo.com)
Seyed Reza Imami (inabim@yahoo.com)

Version: 4 Date: 24 July 2007

Author's response to reviews: see over
Dear The Biomed Central Editorial Team

Thank you very much for your excellent review of our manuscript (Effects of a community-based healthy heart program on increasing healthy women's physical activity: a randomized controlled trial guided by Community-based Participatory Research (CBPR) MS ID: 1363859238124560) and reviewers’ effective comments.

Reviewer (Melissa Napolitano)

Minor Essential Revisions

The requested essential correction in a sentence was done in the discussion.

Best regards,
I. Nabipour, MD